Author’s response to reviews

Title: Profile and outcome of patients with emergency complications of renal failure presenting to an urban emergency department of a tertiary hospital in Tanzania

Authors:

Erasto Sylvanus (kojogare@gmail.com)
Hendry Sawe (hendry_sawe@yahoo.com)
Biita Muhanuzi (biitamuhanuzi@yahoo.com)
Elly Mulesi (ellymulesi@gmail.com)
Juma Mfinanga (jumamfinanga@gmail.com)
Ellen Weber (ellen.weber@me.com)
Said Kilindimo (skilindimo@yahoo.com)

Version: 2 Date: 11 Jan 2019

Author’s response to reviews:

Comment

-- In the section 'Ethics Approval and Consent to Participate', please state whether the informed consent obtained was written or verbal. If verbal, please state the reason and whether the ethics committee approved this procedure.

Response

Written informed consent was sought from all patients, and child assent and parental permission sought prior to enrollment into the study.

Comment

Please do not highlight changes in the text by italicising, underlining, using strikethrough or making the font bold.
Response

Thank you we have uploaded clean manuscript.

Comment

We notice that author JAM is missing from the authors' contributions section. The individual contributions of all authors to the manuscript should be specified in the Authors’ Contributions section. Guidance and criteria for authorship can be found here:

http://www.biomedcentral.com/submissions/editorial-policies#authorship

Response

Thank you, we have rectified this in line with guideline.

Comment

-- At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Response

Thank you, we have uploaded the manuscript